Author’s response to reviews

Title: Levels of oxidative stress biomarkers in seminal plasma and their relationship with seminal parameters.

Authors:

Ali A Khosrowbeygi (khosrowbeygi@yahoo.com)
Nosratollah N Zarghami (zarghami@tbzmed.ac.ir)

Version: 4 Date: 25 May 2007

Author’s response to reviews: see over
Comments for Referees

Title: Levels of oxidative stress biomarkers in seminal plasma and their relationship with seminal parameters.
MS: 6324397831177373

I really thanks in advance for all the reviewers comments and corrections were included as follow:

**Answers to Reviewer: Amalia Carpino**

Major Compulsory Revisions:

2. According to a pilot study of Isoprostane, the minimum sample size was calculated 15 for case and control groups.

English writing was corrected.

**Answers to Reviewer: Maciej Kurpisz**

Major Compulsory Revisions:

The sentences were corrected in the discussion section.

Minor Essential Revisions:

English writing was corrected.

Yours Sincerely
Dr.Zarghami